Author’s response to reviews

Title: The impact of intravenous methylprednisolone pulses on renal survival in anti-neutrophil cytoplasmic antibody associated vasculitis with severe renal injury patients: a retrospective study

Authors:

Yanhong Ma (21518017@zju.edu.cn)
Fei Han (hanf8876@163.com)
Liangliang Chen (llyf@163.com)
Hongya Wang (3100102374@zju.edu.cn)
Haidongqin Han (446752883@qq.com)
Binfeng Yu (1123882010@qq.com)
Ying Xu (xuying7230@163.com)
Jianhuanhui (chenjianghua@zju.edu.cn)

Version: 4 Date: 02 Oct 2017

Author’s response to reviews:

Dear editor,

Thank you very much for reviewing our revised manuscript (BNEP-D-17-00159R3) and giving us the opportunity to revise it. We have studied the comments carefully and have made all the corrections in the manuscript which we hope meet with the approval.

Thank you very much for your attention and consideration. We look forward to hearing from you.

Sincerely,

Fei Han, M.D